Author's response to reviews

Title: Biodegradable collagen matrix (OlogenTM) implant and conjunctival autograft for scleral necrosis after pterygium excision: two case reports

Authors:

Chan-Ho Cho (blitz527@naver.com)
Sang-Bumm Lee (sbummlee@ynu.ac.kr)
Eun-Young Cho (eyecho55@naver.com)

Version: 5 Date: 10 September 2015

Author's response to reviews: see over
Manuscript ID: 7466020091679019

Manuscript Title: Two cases of biodegradable collagen matrix (Ologen®) implant and conjunctival autograft for scleral necrosis after pterygium excision

Authors: Chan-Ho Cho et al.

Dear Reviewers and Editor of BMC ophthalmology(c/c Ms Erica Cruz)

Thank you for your e-mail dated 12 August 2015. We were pleased to know that our manuscript was reviewed as ‘minor essential corrections’.

Based on the instructions provided in your e-mail, we uploaded the file of the revised manuscript in MS word on the journal’s website.

As you notice, we have revised the manuscript by modifying the Abstract, Background, Case presentation, and Conclusion sections based on the comments made by reviewers and editor. Accordingly, we have marked with all the changes made during the revision process. The new or changed text is displayed in red color to first original text and displayed in blue color to last revised text. Appended to this cover letter is our point-by-point response to the comments raised by the reviewers and editor. As you notice, we agreed with all the comments raised by the reviewers and editor. We would like to take this opportunity to express our sincere thanks to the reviewers and editor who identified areas of our manuscript that needed corrections or modification.

We would like also to thank you for allowing us to resubmit a revised copy of the manuscript.

I hope that the revised manuscript will be accepted for publication in BMC ophthalmology.

Sincerely Yours,

Sang-Bumm Lee, MD/PhD
Yeungnam University College of Medicine
Department of Ophthalmology
170, Hyunchung-ro, Nam-gu
Daegu 705-717
South Korea